Author's response to reviews

Title: Case Report: Paravalvular Leak as a Complication of Percutaneous Catheter Ablation for Atrial Fibrillation

Authors:

Orcun Gurbuz (gurbuzorcun@gmail.com)
Abdulkadir Ercan (kadirercan@yahoo.com)
Hakan Ozkan (doctorhakan@hotmail.com)
Gencehan Kumtepe (kumtepe.gencehan@gmail.com)
Ilker H Karal (Ilkerkaral@netscape.net)
Serdar Ener (serdarener@gmail.com)

Version: 2  Date: 20 August 2014

Author's response to reviews: see over
20.08.2014

Dear Editor

On behalf of all the authors, I would like to ask you to reconsider our manuscript entitled “Paravalvular Leak as a Complication of Percutaneous Catheter Ablation For Atrial Fibrillation” for publication in Journal of Cardiothoracic Surgery as a Case Report in Cardiovascular Medicine paper. This report highlights paravalvular leak as a complication of radiofrequency ablation, which is of special interest because these report seems unique in the literature. We conclude all the intended revisions.

We feel that the findings described in this case report will be of special interest to the readers of Journal of Cardiothoracic Surgery.

This manuscript has not been published and is not under consideration for publication elsewhere. All the authors have read the manuscript and have approved this submission. The authors report no conflicts of interest.

Sincerely,

Orcun Gurbuz, MD.

Postal address: Balikesir Universitesi Tip Fakültesi Kalp ve Damar Cerrahisi Anabilim Dalı, Bigadic, 10010, Balikesir, Turkey.

Telephone number: 90-2666121454-1197

fax number : 0266 6121459

e-mail address: gurbuzorcun@gmail.com